Author’s response to reviews

Title: The clinical effect of an unloader brace on patients with osteoarthritis of the knee, a randomized placebo controlled trial with one year follow up.

Authors:

Hjortur Hjartarson (hjorturfr@gmail.com)

Soren Toksvig-Larsen (soren.toksvig-larsen@med.lu.se)

Version: 2 Date: 16 Aug 2018

Author’s response to reviews:

Dear Sirs!

The Manuscript has undergone an extensive English Manuscript Revision by Ms Saskia Titman a native English speaker, healthcare professional and experienced manuscript editor. Multiple minor error have been corrected throughout the document